Author's response to reviews

Title: The Effect of Moderate Physical Activity on Bone Mineral Density and Fracture Rate in Post-menopausal Women aged 75 and over: results from the Canadian Multicentre Osteoporosis Study.

Authors:

Jeffrey M Muir (drjeffmuir@gmail.com)
Chenglin Ye (yecl@mcmaster.ca)
Mohit Bhandari (bhandam@mcmaster.ca)
Jonathan D Adachi (jd.adachi@sympatico.ca)
Lehana Thabane (thabane@mcmaster.ca)

Version: 2 Date: 26 November 2012

Author's response to reviews: see over
November 26, 2012

Editor,
BMC Musculoskeletal Disorders

To the Editor,

Please find attached one (1) copy of a revised manuscript submitted for consideration for publication in BMC Musculoskeletal Disorders. The manuscript, entitled "The Effect of Moderate Physical Activity on Bone Mineral Density and Fracture Rate in Post-menopausal Women aged 75 and over: results from the Canadian Multicentre Osteoporosis Study", carries the identification number: MS: 9810954448523464. The following revisions, as requested, have been made:

§ A statement regarding ethics review board approval has been added to the Methods section,
§ A statement regarding informed consent from each participant has been added to the Methods section,
§ Author’s contributions have been expanded upon,
§ Funding sources and competing interest statements have been expanded upon,
§ The acknowledgements section has been expanded to include reference to the CaMos group and it’s specific members, and
§ Figure 1 has been removed from the text and uploaded separately.

Thank you for your consideration. Please feel free to contact me with any additional questions or concerns.

Best regards,

Dr. Jeff Muir, MSc, DC, MSc (Clin Epi)
Manual Therapy & Rehabilitation
Clinical Epidemiologist
Department of Clinical Epidemiology & Biostatistics
McMaster University, Hamilton, ON
c/o 18-550 Fennell Ave. East,
Hamilton, ON L8V 4S9